Author's response to reviews

Title: Comparison of the Glidescope, flexible fibreoptic intubating bronchoscope, iPhone modified bronchoscope, and the Macintosh laryngoscope in normal and difficult airways: A manikin study.

Authors:

Adrian Langley (adrianlangley@gmail.com)
Gabriel Mar Fan (Gabriel_Marfan@health.qld.gov.au)

Version: 3
Date: 17 February 2014

Author's response to reviews: see over
Dear Editor

Please find a new draft uploaded in which I have added statistics and numbers to the abstract and altered the conclusion sections to represent what I think is the main result of the study.

Sincerely

Dr Adrian Langley

Editor's comment

Title: Comparison of the Glidescope, flexible fibreoptic intubating bronchoscope, iPhone modified bronchoscope, and the Macintosh laryngoscope in normal and difficult airways: A manikin study.

Authors: Adrian Langley (adrianlangley@gmail.com) Gabriel Mar Fan (Gabriel_Marfan@health.qld.gov.au)

Version: 2 Date: 23 January 2014

Editor's comment:
1. Please, add numbers and statistics to the abstract to characterize key results of the manuscript.
2. Modify the current Conclusion section, and present the effective main conclusions of the papers, based on the main results of the study.